Author’s response to reviews

Title: Papillary Thyroid Carcinoma Presented as a Primary Renal Tumor with Multiple Pulmonary and Bone Metastases: A Case Report

Authors:

Emre Gezer (gezer_emre@hotmail.com)
Alev Selek (alevselek@gmail.com)
İlhan Tarkun (ilhantarkun@superonline.com)
Zeynep Cantürk (zeynepcanturk@hotmail.com)
Berrin Çetinarslan (barslan@kocaeli.edu.tr)

Version: 1 Date: 30 Nov 2018

Author’s response to reviews:

to Reviewer #2:

The variant of papillary thyroid carcinoma cannot be specified, because there is no detail about it on her handwritten patient folder. Additionally, since electronic archive system at our clinic is not older than 10 years, the original pathological report cannot be provided.